

G OPEN ACCESS

Citation: Holmen S, Galappaththi-Arachchige HN, Kleppa E, Pillay P, Naicker T, Taylor M, et al. (2016) Characteristics of Blood Vessels in Female Genital Schistosomiasis: Paving the Way for Objective Diagnostics at the Point of Care. PLoS Negl Trop Dis 10(4): e0004628. doi:10.1371/journal.pntd.0004628

Editor: Ricardo Toshio Fujiwara, Universidade Federal de Minas Gerais, BRAZIL

Received: July 21, 2015

Accepted: March 23, 2016

Published: April 13, 2016

Copyright: © 2016 Holmen et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: The project received funding from the European Research Council under the European Union's Seventh Framework Programme, the University of Copenhagen with the support from the Bill and Melinda Gates Foundation, the Norwegian Research Council, the South-Eastern Regional Health Authority of Norway, the University of Oslo, Norway and Oslo University Hospital, Norway. The funders had no role in study design, data collection **RESEARCH ARTICLE**

Characteristics of Blood Vessels in Female Genital Schistosomiasis: Paving the Way for Objective Diagnostics at the Point of Care

Sigve Holmen^{1,2}*, Hashini Nilushika Galappaththi-Arachchige^{1,2}, Elisabeth Kleppa^{1,2}, Pavitra Pillay³, Thajasvarie Naicker⁴, Myra Taylor⁵, Mathias Onsrud⁶, Eyrun Floerecke Kjetland^{1,5}, Fritz Albregtsen^{7,8}

 Norwegian Centre for Imported and Tropical Diseases, Department of Infectious Diseases Ullevaal, Oslo University Hospital, Oslo, Norway, 2 Institute of Clinical Medicine, University of Oslo, Oslo, Norway,
Department of Biomedical and Clinical Technology, Durban University of Technology, Durban, South Africa, 4 Optics and Imaging Centre, Nelson R Mandela School of Medicine, College of Health Sciences, University of KwaZulu-Natal, Durban, South Africa, 5 Discipline of Public Health Medicine, Nelson R Mandela School of Medicine, College of Health Sciences, University of KwaZulu-Natal, Durban, South Africa,
Department of Gynaecology, Oslo University Hospital, Oslo, Norway, 7 Department of Informatics, University of Oslo, Oslo, Norway, 8 Institute for Cancer Genetics and Informatics, Oslo University Hospital, Oslo, Norway

* sigve.holmen@gmail.com

Abstract

Background

The mucosal changes associated with female genital schistosomiasis (FGS) encompass abnormal blood vessels. These have been described as circular, reticular, branched, convoluted and having uneven calibre. However, these characteristics are subjective descriptions and it has not been explored which of them are specific to FGS.

Methods

In colposcopic images of young women from a schistosomiasis endemic area, we performed computerised morphologic analyses of the cervical vasculature appearing on the mucosal surface. Study participants where the cervix was classified as normal served as negative controls, women with clinically diagnosed FGS and presence of typical abnormal blood vessels visible on the cervical surface served as positive cases. We also included women with cervical inflammatory conditions for reasons other than schistosomiasis. By automating morphological analyses, we explored circular configurations, vascular density, fractal dimensions and fractal lacunarity as parameters of interest.

Results

We found that the blood vessels typical of FGS are characterised by the presence of circular configurations (p < 0.001), increased vascular density (p = 0.015) and increased local connected fractal dimensions (p = 0.071). Using these features, we were able to correctly classify 78% of the FGS-positive cases with an accuracy of 80%.

and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

Conclusions

The blood vessels typical of FGS have circular configurations, increased vascular density and increased local connected fractal dimensions. These specific morphological features could be used diagnostically. Combined with colourimetric analyses, this represents a step towards making a diagnostic tool for FGS based on computerised image analysis.

Author Summary

Female genital schistosomiasis (FGS) is a disease of the female genitalia caused by a waterborne parasite. The parasite lays eggs that may be found in the genital organs and these eggs cause local inflammation and subsequent lesions. These lesions are called sandy patches and they are often accompanied by abnormal blood vessels. Sandy patches can be hard to see and clinicians have suggested that the typical blood vessels may serve as supporting or independent diagnostic indicators of FGS. Clinicians have described the specific features of the vessels as: appearing as circular shapes, forming a network and having branches. In this study, we examined the shape and appearance of cervical blood vessels by computer image analysis. We found that the blood vessels in women with FGS appear more often as circular shapes, at higher density and with more complex shapes compared to those in healthy women and in women with inflammation of the cervix for other reasons than FGS. Our findings suggest that cervical vessel appearance can contribute to the diagnosis of FGS. It could also be used in a future diagnostic tool using computer analysis. Further research is needed to determine if FGS can be diagnosed objectively.

Introduction

The parasite *Schistosoma* (*S.*) *haematobium* deposits eggs in the urogenital tract causing urogenital schistosomiasis, and in females, it may cause the clinical syndrome known as female genital schistosomiasis (FGS) [1]. Studies have shown that 58-75% of women with detectable excretion of *S. haematobium* eggs in urine may have genital manifestations [1,2]. Likewise, in endemic areas, as many as 41% of women without detectable eggs in urine have been found to have genital lesions [1]. Genital lesions associated with *S. haematobium* ova are observed as grainy sandy patches, homogenous yellow sandy patches and rubbery papules [3,4]. In addition there is often the presence of abnormal blood vessels and occasionally these blood vessels are seen alone [1,3,4]. It has been hypothesized that the increased vascularity represents neovascularization [5]. However, a histopathological study on cervical biopsies did not find active neovascularization around schistosome eggs [6]. One case report presenting clinical images alongside the histological correlates showed dilated and tortuous venules containing viable schistosome eggs surrounded by a thrombus [7]. Clinically, these abnormal blood vessels have been described as being circular, reticulated, branched, convoluted and of uneven calibre [1,3,4].

FGS causes genital bleeding, pelvic discomfort and infertility [3,8]. There is also cross-sectional evidence that urogenital schistosomiasis increases the odds of having human immunodeficiency virus (HIV) by 2.9–4.0 times [9–11]. The diagnosis of FGS is based on the clinical identification of the characteristic genital lesions and abnormal blood vessels by visual inspection [12]. In areas where urogenital schistosomiasis is co-endemic with HIV, taking biopsies is of ethical concern due to the unnecessary risk of HIV transmission imposed on the patient by the iatrogenic lesion [3]. Urinary and genital egg excretion are poor proxy markers of lesions and systemic antigen / antibody tests do not provide information on the anatomical site of the morbidity [3]. Therefore, there is no adequate objective point-of-care diagnosis for FGS, making it necessary to explore alternative diagnostic tools to support the clinical diagnosis.

A safe, simple and reliable diagnosis for FGS is essential to provide patients with proper care and insight into this chronic condition [3]. Treatment for the active infection must be sought but the patient must also be made aware of the chronic nature of the lesions and the potential higher susceptibility to HIV. For many patients, it will be of enormous value to have an explanation for their genital symptoms and this may prevent repeated and unnecessary treatments for STIs based on a syndromic diagnosis.

Telemedicine is an approach that currently brings healthcare expertise to the point-of-care in the fields of pathology, dermatology and radiology [13]. We have previously suggested that the emerging smart phone availability in developing countries may be used as a platform to capture and interpret the characteristic sandy patches with computerized colour analyses as a possible approach to a simple diagnosis at the point of care [14,15]. Morphological vessel analysis may represent an additional approach to the diagnostic problem, either by itself or as a supplement to the colourimetric analysis. Such a diagnostic tool could be made freely available and easy to distribute to remote, rural areas as a software application, e.g. for smartphones or laptops. Many health clinics in low-resource settings already use digital cameras in their cervical cancer screening programs [16,17]. In these clinics it would not represent any additional cost to implement a free, software-based diagnostic tool for FGS detection.

We hypothesize that computerized morphological analyses may be used to objectively identify and quantify the abnormal blood vessels associated with FGS. In this study we explore the morphological features of cervical blood vessels observed in three groups: (1) in cases with FGS, (2) in endemic controls with other genital pathology and (3) in healthy endemic controls.

Methods

Ethics

The study was granted permission by the European Group on Ethics in Science and New Technologies (2011, Ref: IRSES-2010:269245), the Biomedical Research Ethics Administration, University of KwaZulu-Natal (March 2012, Ref: BF029/07), the regional Department of Health (DOH), Pietermaritzburg, KwaZulu-Natal (February 2009, Ref: HRKM010-08) and the Norwegian ethics committee, REC South East (2007 and 2011, Ref: 469-07066a1.2007.535), and the district Departments of Health and Education, KwaZulu-Natal in September 2012 and November 2012, respectively.

Participant details were kept confidential and each was allocated a unique number. All colposcopic images were non-identifiable: They only depict the uterine cervix and contain no names. All participants signed written, informed consent forms prior to investigation and were informed of the right to withdraw at any moment if they so wished. All participants were tested for human immunodeficiency virus, abnormal cytology and sexually transmitted infections. Test results were given to study participants who wanted their results. Counselling, referral and treatment were in accordance with standard DOH, South African guidelines. Anti retroviral therapy is free of charge in South Africa.

Study population and area

The study participants were recruited as part of a larger study exploring FGS in young women in South Africa between 2011–2013. Participants were recruited from rural schools north and south of Durban, KwaZulu-Natal. These are areas endemic for urogenital schistosomiasis [18].

Sexually active, non-pregnant young women aged 16 and above were recruited. Recruitment was not based on symptoms or test results.

Clinical investigation and questionnaire

Trained research assistants interviewed all participants in the local language (isiZulu). The questionnaire included questions on sexual behaviour, pregnancy and contraception. Gynaecological investigation was performed in those who consented. All clinical findings were recorded in electronic format along with a schematic representation of the lesion appearance, size and location (if present). As reported previously, images were captured colposcopically [14,15]. We searched the database (n = 1715) for images that fulfilled the following criteria: the cervix should be in the field of view, there should be no foreign material in the field of view (swab, spatula, acetic acid etc.) and the exposure and focus should be adequate for visualisation of the cervical surface and its features.

We defined three groups based on the clinician's evaluation and subsequent laboratory findings: (1) women with a normal appearing cervix and no schistosome egg excretion in urine, (2) women with abnormal blood vessels typical of FGS who had egg excretion in urine or presence of sandy patches and (3) women with a cervix presenting signs of inflammation (oedema, swelling and / or rubor) but no clinical findings typical of FGS and no schistosome egg excretion in urine. Each group consisted of randomly selected participants from the database.

Laboratory analyses

A single urine sample was collected between 10 a.m. and 2 p.m. on the day of the clinical examination. Merthiolate-formalin solution (2%) was added to 10 mL of the sample. The sample was centrifuged and the pellet was deposited on two slides. Two independent technicians screened each of the slides by light microscopy [19].

Traditional Pap smears were done by the investigating clinician, preserved with a commercial cytological spray-fixative, then stained and analysed by cytotechnologists. Smears were reported using the Bethesda System of Reporting [20], and the categories of atypia included. Cellular atypia was classified as atypical squamous cells of unknown significance (ASCUS), low- and high-grade squamous cell intraepithelial lesions (LSIL and HSIL). The number of neutrophils and degree of inflammation were graded as none, mild, moderate or marked. A syndromic protocol was used to diagnose and treat findings at the point of investigation in alignment with the practice in rural clinics. Patients were contacted and asked if they had been treated and helped with further management of the disease once laboratory results were available. Patients with cellular atypia were referred to their local hospital for further management.

Cervico-vaginal lavage (CVL) was collected by spraying 10 mL of saline four times onto the ectocervical mucosa followed by withdrawal back into a syringe. The CVL was analysed for *Trichomonas vaginalis* using an in-house real-time PCR technique (Laboratory of Infection, Prevention and Control, University of KwaZulu-Natal, Durban, South Africa). *Chlamydia (C.) trachomatis* and *Neisseria gonorrhoea* were analysed using strand displacement assay on a ProbeTec machine (Becton, Dickinson and Company [BD], Franklin Lakes, New Jersey, USA). The CVL was also centrifuged, smeared on a slide and scored using the Nugent's criteria for bacterial vaginosis.

Treponema pallidum was analysed using rapid plasma reagin (RPR) on serum (Macro Vue 110, BD, Franklin Lakes, New Jersey, USA) and positive samples were confirmed using treponema pallidum haemagluttination assay (TPHA, Immutrep, Omega diagnostics Group PLC, Alva, United Kingdom). *Herpes simplex* type 2 antibodies were detected in serum using ELISA (Ridascreen HSV 2 IgG, Davies Diagnostics, Randburg, South Africa). HIV was detected in serum using a rapid antibody test (Core One Step HIV 1/2 test kit, Kendon laboratories, Durban, South Africa). Positive tests were confirmed using a different rapid antibody test (Sensa Tri-Line HIV Test Kit, Pantech (Pty) Ltd, Durban, South Africa).

Blood vessel analyses

All image processing and analyses were performed using ImageJ version 1.49 (open source, free software from National Institutes of Health, US). Plugins for ImageJ were written in Java (Oracle Corporation, Redwood Shores, US) to perform the specific analyses. A macro was written to automate the execution of all the analyses and to record all the results. Numeric results were recorded in a text file and images were generated to allow for visual verification of the analyses (an example is given in Fig 1F).

Colposcopic images often contain non-mucosal elements such as parts of the speculum, medical instruments and skin that should not be subject to the image analysis. As reported previously, we therefore applied automated detection of the region of interest (ROI) prior to analysis, using a previously described method, which identifies the cervix as the central area in the image having the highest values in the "a-channel" (of the "Lab colour space") [14].

In order to do morphological analyses on the cervical blood vessels, the blood vessel structures were identified by splitting the original colour image into the "green channel" (of the "RGB colour space") and the "saturation channel" (of the "HSV colour space"). The blood vessels, which appear bright red, have very low values of green and will therefore appear dark in the green channel. Furthermore, the blood vessels have high values of saturation and will therefore appear bright in the saturation channel. However, the ectocervix is convex and the surface is not evenly illuminated by the colposcope. It was therefore necessary to equalize the green and saturation channels. By generating an inverted image, which was subsequently smoothed using a Gaussian blur filter and then calculating the sum (addition) with the original image, the irregular illumination of the ectocervix was eliminated.

The product (multiplication) of the inverted green channel and the saturation channel resulted in an image where the blood vessels had higher values than the initial images (Fig 1B and S3 Fig). This resulting image was used for the subsequent analyses and is henceforth referred to as the processed image.

Morphological analyses

Template matching allows for the identification of structures resembling a pre-defined template [21]. In ophthalmology, template matching algorithms have been developed to identify the optical nerve [21,22]. A circular template was generated to resemble the characteristic circular shape of the vessels as indicated by experienced clinicians (S6C Fig). For the template matching, the vascular structures were isolated from the processed image by removing all pixels below the mean value of grey (Fig 1C and S6B Fig). Template matching by convolution is a very processor-intensive process if applied as a pixel-by-pixel approach in the image domain, which would be impractical for a cell-phone application. Therefore, template matching was performed by converting the template and the image to the frequency domain by the fast Fourier transform [23], multiplying them and finally converting the result back to the image domain (multiplication in the frequency domain corresponds to convolution in the image domain). In the resulting image, areas with circular configurations have higher intensity (Fig 1D and S6D Fig). A threshold level was set to remove pixels below the 97.5th percentile of intensity (S6E Fig). The final image contains clustered pixels representing the centres of the matched circular structures (Fig 1D and S6F Fig). The number of circles identified was recorded for each image.



Fig 1. Image analysis of female genital schistosomiasis of the uterine cervix. A. The original colour image. Note that there is a light reflection at 6 o'clock and parts of the speculum can be seen on the right hand side (artefacts). These were removed automatically in the image processing. B. The product of multiplying the inverted "green channel" (from the "Red-Green-Blue" (RGB) colour space with the "saturation channel" (from the "Hue-Saturation-Value" (HSV) colour space). C. The region of interest (ROI) of image B with all pixels below the mean grey value removed. D. The result of convolution of the circular template and image C. The darker areas represent higher degrees of roundness. E. The result of adaptive local thresholding of image B using a modified Niblack method. F. The final output image automatically generated by the image analysis. It shows the ectocervix with numerous abnormal blood vessels (black skeleton) and the centres of circular structures identified by template matching (yellow dots).

doi:10.1371/journal.pntd.0004628.g001

For the remaining morphological analyses, the processed image was converted to a binary image by an algorithm based on the Niblack [24] method for local adaptive thresholding, where the local threshold $T(x, y) = \mu(x, y) + k * \sigma(x, y)$, where μ and σ are the local mean value and standard deviation, respectively, within a sliding, circular window with a 50 pixel radius. Instead of a constant k-value determined empirically by trial-and-error [25] (S4E Fig), we inferred the optimal k-value for each image by first estimating the most likely distributions of foreground and background by using an expectation-maximisation (EM) algorithm (k-means clustering) (S4F Fig).

However, local adaptive thresholding may result in noise in areas with low contrast. We therefore only applied the Niblack threshold if the local standard deviation exceeded the mean standard deviation of all possible windows. If it did not, the area was defined as "low-contrast", and the threshold was set to the most likely foreground grey value (as estimated by the EM-algorithm). Furthermore, local adaptive thresholding may produce perimeter artefacts when applied on an isolated structure laid over a background, due to artificially high contrast between background and structure (S4E Fig). This was the case in our approach since we performed all analyses within a ROI. We therefore also used the most likely foreground grey value as threshold value for pixels whose window (50-pixel radius) fell within the perimeter of the ROI. Noise was removed from the resulting binary image by using a median filter with a radius of 4 pixels. An example of a resulting binary image is shown in Fig 1E.

After smoothing, the structures were skeletonised to a single pixel width, since in this context we are interested in the spatial configuration of the vessels. The skeletonisation was done using a lookup table to repeatedly remove pixels from the edges of objects in the binary image, analysing 3x3 pixel grids [26].

The skeletonised images were first used for calculating the distance between vessels and the total number of vessels per image. Distinct vessels were defined as not being connected by an 8-neighbourhood relationship. The distance between blood vessels was defined as the distance from any given vessel to its' closest neighbour in a straight line. The skeletonised blood vessels were superimposed on the original colour image (Fig 1F) to allow for visual verification by the clinician. Experienced clinicians confirmed that the patterns portrayed in Fig 1F are recognized as typical in FGS cases (personal communication, EFK and HNGA).

Blood vessels may be considered fractals; objects whose details under magnification resemble the structure as a whole (S1 Text and S1 Fig and S2 Fig) [27]. The calculated fractal dimension of a network of vessels increases with increasing complexity [27]. A single point has a fractal dimension of zero, a straight line has a fractal dimension of one and a plane has a fractal dimension of two. However, a convoluted vessel in a plane will have non-integer fractal dimension (D) between that of a line (D = 1.0) and a plane (D = 2.0). Analysis of local connected fractal dimensions (LCFD) allows for identification of areas with higher or lower fractal dimensions within a fractal [27]. In ophthalmology, it was possible to identify patients with occlusion of the retinal artery by analysis of LCFD [27]. Another study found that patients with cerebral lacunar strokes had lower fractal dimensions of the retinal vessels compared to patients with minor cortical strokes [28].

Fractal lacunarity is a counterpart to the fractal dimension describing the heterogeneity of a fractal in terms of the size and distribution of holes or gaps [29]. If a fractal has large gaps or holes, it has high lacunarity. In gynaecology, fractal dimension and fractal lacunarity have been used to classify the uterine vessels seen in hysteroscopic images of patients with endometrial cancer and abnormal uterine bleeding [30].

We used the box counting method [29], with boxes doubling in size from 2–128 pixels, to estimate the general fractal dimension and lacunarity for the vascular network (<u>S5 Fig</u>). The LCFDs were calculated using the free ImageJ plugin FracLac version 2015Marb6206 (Charles Sturt University, New South Wales, Australia).

Statistical analyses

Statistical analyses and graphs were produced using IBM SPSS Statistics Version 19 (IBM Company, Chicago, USA).

A sample size calculation showed that we needed a minimum of 36 cases in each group in order to be able to demonstrate a difference of 10% or more within a 95% interval of confidence if the standard deviation is 15 or less (arbitrary units). We decided to include 50 cases in each of the three groups.

Group characteristics were compared using the Kruskal-Wallis analysis of variance (age, days since last menstrual period and days since last intercourse) and the Chi-square test (pregnancy, hormonal contraception, STIs and findings in Pap-smear).

Comparisons of morphological characteristics were done using bivariate logistic regression; comparing the normal cervical appearance to those with blood vessels typical of FGS and those with cervical inflammation.

A multivariable logistic regression model was constructed using all the morphological characteristics and the group characteristics that differed significantly between the groups within an 85% interval of confidence. Variables were eliminated from the model one by one (backwards elimination) based on a minimum significance level of 0.15 and only if the model's likelihood ratio did not change significantly (p < 0.05). The final regression model was used to construct a ROC-curve to find the optimal cut-off value for identifying images with abnormal blood vessels typical of FGS, by identifying the point of the curve closest to the upper left corner. Classification accuracy was calculated using the cut-off value found on the ROC-curve.

Results

Group characteristics

The group characteristics are presented in <u>Table 1</u>. The women were similar in all respects except for the use of hormonal contraceptives, the prevalence of *C. trachomatis* and *Herpes simplex*. These variables were therefore considered when constructing the multivariable regression model. The morphological analyses presented in <u>Table 2</u> show that women with FGS have significantly more circular blood vessels and a higher density of vessels are visible on the surface. In addition the fractal dimensions were higher in women with FGS.

In women with FGS, we found that the strongest morphological predictor of pathology was the presence of circular configurations in the vascular network (<u>Table 3</u>). We also found increased vascular density and increased vessel complexity (as estimated by LCFD) in women with FGS (Tables <u>2</u> and <u>3</u> and <u>Fig 2</u>).

In the multivariable analysis, the number of blood vessels, general fractal dimension, general fractal lacunarity and peak LCFD were eliminated from the regression model along with the possible confounders; hormonal contraceptives, *C. trachomatis* and *Herpes simplex* (Table 3). The final regression model was used to generate a ROC curve (Fig 3) and the optimal cut-off

Variable	Abnormal blood vessels typical of FGS (n = 50)	Cervical pathology without signs of FGS (n = 50)	Normal cervical appearance (n = 50)	P-value of group difference
Demographics and behaviour				
Median age (range)	19 (16–26)	19 (16–26)	19 (16–27)	0.453 ^a
Uses hormonal contraceptives	32%	56%	41%	0.049 ^b
Has been pregnant	46%	60%	58%	0.315 ^b
Median number of days since last period	17	15	15	0.581 ^a
Median number of days since last intercourse	38	18	20	0.335 ^a
Median age at sexual debut	16.5	17	16	0.856 ^a
Sexually transmitted infections				
Chlamydia trachomatis	19%	40%	29%	0.079 ^b
Herpes simplex	28%	45%	28%	0.141 ^b
HIV	24%	34%	18%	0.190 ^b
Trichomonas vaginalis	18%	31%	17%	0.699 ^b
Neisseria gonorrhoea	4%	8%	6%	0.183 ^b
Treponema pallidum	0%	2%	2%	0.589 ^b
Median Nugent's criteria	7	8	8	0.237 ^a
Findings in Pap-smear				
ASCUS ^c	25.9%	24.4%	24.0%	0.985 ^b
LSIL ^d	37.0%	17.1%	28.0%	0.176 ^b
HSIL ^e	3.7%	4.9%	0%	0.546 ^b
Marked inflammation in Pap- smear	80%	82%	63%	0.182 ^b
Marked neutrophilia in Pap- smear	60%	53%	36%	0.217 ^b

Table 1. Group characteristics.

a Kruskal-Wallis analysis of variance

b Chi-square test

c Atypical squamous cells of unknown significance

d Low-grade squamous cell intraepithelial lesion

e High-grade squamous cell intraepithelial lesion

P-values in bold indicate variables considered for the multivariable regression model (inclusion criteria p < 0.15)

doi:10.1371/journal.pntd.0004628.t001

value was -0.037. Using this cut-off value for the regression model, the model's classification accuracy was calculated (<u>Table 4</u>). Finally, by defining the images with blood vessels typical of FGS as positive cases and the combined set of images with normal cervical appearance and cervical inflammation as negatives, the overall ability to classify the image correctly was 78% for positive cases and 80% for negative cases.

Discussion

In this study we have demonstrated that computerised, morphological image analyses can be used to distinguish blood vessels typical of FGS from normal vessels and from blood vessels on cervices with signs of inflammation caused by other diseases. Using the morphological feature of circularity (as determined by template matching), distance between vessels and LCFD, we were able to classify the images with an accuracy of 78% for positive cases and 80% for negative cases.

Table 2. Characteristics of the blood vessels.

	Abnormal blood vessels typical of FGS (n = 50)	Cervical pathology ^a without signs of FGS (n = 50)	Normal cervical appearance (n = 50)
	Median (range)	Median (range)	Median (range)
Circularity			
Number of circular template matches ^a	20.5 (1–63) **	9.0 (0–29)	10.5 (0–23)
	OR 1.12 for an increase of 1 template match (95% CI: 1.08–1.23)		
Vascular density			
Number of blood vessels ^b	109 (51–238) **	82.5 (6–174)	72 (6–123)
	OR 1.29 for an increase of 10 blood vessels (95% CI: 1.13–1.47)		
Distance between blood vessels (px)	62.0 (49.9–98.2) *	71.9 (48.9–720.1)	84.0 (49.8–811.1)
	OR 0.96 for an increase of 1 px (95% CI: 0.93–0.99)		
Fractal dimensions			
Fractal dimension ^d	1.029 (0.938–1.115) *	0.987 (0.817–1.074)	1.011 (0.820–1.086)
	OR 1.16 for an increase of 0.01 dimensions (95% CI: 1.04–1.29)		
Fractal lacunarity ^e	0.206 (0.160-0.288) *	0.231 (0.180–0.344) *	0.221 (0.173–0.280)
	OR 0.83 for an increase of 0.01 in lacunarity (95% CI: 0.70–0.98)	OR 1.21 for an increase of 0.01 in lacunarity (95% CI: 1.04–1.42)	
Mean local connected fractal dimension	1.011 (0.978–1.035)	1.010 (0.954–1.028)	1.008 (0.970–1.032)
Peak local connected fractal dimension	1.501 (1.395–1.564) *	1.485 (1.265–1.563)	1.477 (1.268–1.580)
	OR 1.17 for an increase of 0.01 dimensions (95% CI: 1.05–1.31)		

a. Defined by the clinician as oedema, swelling and / or rubor.

b. Presence of configurations matching a predefined circular template

c. Absolute number of vessels per image

d. Structural complexity

e. Structural heterogeneity

* Results differ significantly from the normal cervical appearance, evaluated using bivariate logistic regression, p < 0.05

** Results differ significantly from the normal cervical appearance, evaluated using bivariate logistic regression, p < 0.001

OR (Odds ratio) is calculated in reference to the normal cervical appearance

doi:10.1371/journal.pntd.0004628.t002

Table 3. A multivariable regression model describing the association between abnormal blood ves
sels typical of FGS and morphological vessel characteristics.

Variable	Coefficient ^a	Odds ratio (95% CI)	p-value
Circular template matches	0.147	1.16 (1.08–1.25)	< 0.001
Mean distance between vessels	-0.042	0.96 (0.93–0.99)	0.015
Mean local connected fractal dimension	-0.531	0.590 (0.33–1.05)	0.071
Constant	54.464		

a. The regression coefficients were used to calculate the ROC curve (Fig 3) and the optimal cut-off value for a positive FGS diagnosis.

doi:10.1371/journal.pntd.0004628.t003

PLOS | NEGLECTED TROPICAL DISEASES



- Normal cervical appearance
- Abnormal blood vessels typical of FGS
- Cervial pathology without signs of FGS

Fig 2. Scatterplots showing the co-variation of variables included in the regression model. The distribution of values of each of the variables included in the regression model differ between the women with FGS and the two other groups: The number of circular template matches is higher, the distance between vessels is lower and the mean local connected fractal dimension tends to be lower, although the latter is not significant.

doi:10.1371/journal.pntd.0004628.g002

Clinical applicability

It has been reported that abnormal blood vessels typical of FGS present with specific morphological features [3]. Our findings support this observation and confirm that assessment of vessel appearance is of clinical significance when diagnosing FGS. Furthermore, our findings suggest that computerised image analyses of blood vessels visible on the cervical surface may play a role in developing a diagnostic tool for FGS, possibly together with colour analyses of sandy patches [14,15]. The World Health Organization recommends criteria for point-of-care tests: Affordable, Sensitive, Specific, User-friendly, Rapid and robust, Equipment-free and Deliverable to end-users (ASSURED) [31-33]. No such diagnostic tools exists for FGS and, to our knowledge, none are in the pipeline. Cellular phones are increasingly available in schistosomiasis endemic areas and a software-based diagnostic tool can be made freely available (for download) for use on existing devices such as smartphones, tablets or laptops. The software can be designed in a user friendly way and it can analyse the images instantly on the device, at the point of care (< 10 seconds using current laptops). However, it needs to be verified whether such a method is sufficiently accurate and robust when applied in a low-resource setting by local health professionals.

All the image analyses performed by this research group have been done using the Javaframework provided by ImageJ, which is open source and available on a number of platforms. It can therefore easily be deployed on any unit capable of running ImageJ. For implementation on Android devices, the necessary Java-libraries would simply need to be bundled in the application package. For other devices (not running Java), implementation might prove to be more laborious as the methods might need to be rewritten to conform to the operating system's base language.

The current paper explores the possibility of using the morphological features of the abnormal blood vessels associated with FGS but this research group has previously published methods for diagnostics based on image analysis using colour to detect sandy patches [14,15]. The significance of these various lesions in terms of the pathophysiology and morbidity of FGS is not well understood. They may represent different aspects of the disease, different degrees of severity or progression. A diagnostic algorithm should include all the characteristic lesions and



Fig 3. Receiver operating characteristics curve for the diagnosis of FGS by blood vessel analysis. The curve was generated using the regression model presented in Table 3. Area under the curve = 0.872.

doi:10.1371/journal.pntd.0004628.g003

future studies may try to decipher their individual significance. Combined, they might also provide increased diagnostic accuracy [15].

We found a significantly higher number of circular features in the vascular network of patients with FGS. Furthermore, we found decreased distance between vessels and increased LCFD in women with FGS, indicating that the cervical vascular network is denser and morphologically more complex than in other diseases or in normal cervices. Increased LCFD is a morphological feature that requires computer analysis and will therefore only be clinically relevant when the diagnostic software has been introduced. However the two other findings (circular

	Table 4.	Classification of	f images using	g the proposed	d regression mod	el
--	----------	-------------------	----------------	----------------	------------------	----

Regression model ^a	FGS	Other cervical pathology	Normal cervix
Blood vessel ^b negative	11 (22%)	38 (76%)	42 (84%)
Blood vessel ^b positive	39 (78%)	12 (24%)	8 (16%)

^a Using a cut-off value of -0.037 for a positive diagnosis

^b Typical for FGS (circular, high density, with high fractal dimensions)

doi:10.1371/journal.pntd.0004628.t004

vessels and increased vessel density) can be recognised by a trained clinician and therefore represent objective, clinical observations that complement and support diagnosis of FGS that is based on inspection.

Pathophysiological hypotheses

The circular vessel configuration which is seen more frequently on cervices of women with FGS could represent an area where an egg granuloma acts as a foreign body that locally prevents the growth of blood vessels or pushes them aside as the granuloma grows. Alternatively, the occlusion of a blood vessel by a schistosome worm-pair; its eggs and/or egg-induced thrombosis could cause dilatation and varicose-like distortion of the vessel [7].

In ophthalmology, the retinal vasculature has areas with increased LCFDs in patients with occlusion of the central vein or retinal artery [27]. Our finding of increased LCFDs in women with FGS therefore supports the argument that occlusion of small vessels may in fact represent an important feature in urogenital schistosomiasis.

Limitations

We were not able to assess vessel calibre, as the images did not provide sufficient detail for this. Uneven calibre has been proposed as one of the characteristic features of FGS. However, this is difficult to assess without high magnification colposcopy and may therefore not be suitable for a low-cost, simple diagnostic approach.

Furthermore, the analyses in this study were performed on images acquired colposcopically. However, colposcopes are generally not available in clinics in endemic areas [34]. Previous attempts at performing computerised colour analyses on images of simulated low quality have shown good results [14]. Simple, handheld cameras have been evaluated for use in visual inspection with acetic acid (VIA) in cervical cancer screening programmes [35,36]. Similarly, this method should be evaluated on images acquired using simple devices such as handheld cameras or mobile devices. One obstacle that will need to be addressed when using simple handheld devices, is even illumination of the field. This is provided when using a colposcope, but for handheld devices it might be necessary to use an external light source such as a flash-light or a lamp. Furthermore, where colposcopes are available, these are primarily used for cancer screening. It would therefore be prudent not to depend on such equipment for the diagnosis of FGS, as there could be a false reassurance in regards to excluding cancer as a differential diagnosis if the colposcope is sometimes used for the diagnosis of FGS without targeted screening for cancerous lesions.

The analyses of colposcopic images are also limited by the narrow depth of field (the area in focus) provided by the colposcope. Since the ectocervix is generally convex in shape, only parts of its surface can be within the depth of field at once when photographed by a colposcope. Furthermore, it is rarely possible to visualise all the surfaces of the lower genital tract, especially the fornices and the vaginal wall. This may result in false negative cases in which lesions are missed. However, this would skew our results towards the null-hypothesis, thus strengthening our findings.

The participants in this study are young women aged 16–27 and cervical appearance may differ in older women due to transformation of the ectocervical columnar epithelium by metaplasia into squamous epithelium (the transformation zone) with increasing age [37].

Conclusion

Blood vessels typical of FGS appear in circular configurations, have increased density and increased local connected fractal dimensions (LCFD). Future studies should assess these findings alongside the colourimetric analyses of the sandy patches in images acquired using a

simple digital camera. Furthermore, the digital application should be explored as a supplement to the visual inspection.

Supporting Information

S1 Text. Fractals and blood vessel morphology. Details about fractals and how we analyses blood vessel morphology using fractal properties. (DOCX)

S1 Fig. South Africa's coastline can be considered a fractal. South Africa's coast line reveals more details when visualised at increasing magnifications (from left to right) but the structures are fundamentally similar.

(TIF)

S2 Fig. Measuring the outline of South Africa. The outline of South Africa can be measured with increasing levels of detail (number of line segments), revealing a more complex structure and also adding to the length of the outline. (TIF)

S3 Fig. Combining the green and saturation channels. The multiplication of the inverted green channel and the saturation channel results in an image where the blood vessels have "boosted" values, appearing more clearly than in either of the two original colour channels. (TIF)

S4 Fig. Thresholding. A. Original colour image. **B.** Equalized product of the inverted Green channel and the Saturation channel. The automatically detected region of interest (ROI) is indicated in red. **C.** The standard deviation of grey values in a 50px window calculated for each pixel and represented as relative levels of grey values (0–255). **D.** Thresholding performed using the mean grey value. **E.** Thresholding using the standard Niblack method with a k-value of -0.2. Notice the perimeter artefact as a thick white line around the ROI. **F.** Thresholding using Niblack's method with calculation of optimal k-value per 50px sliding window and elimination of the perimeter effect. (TIF)

S5 Fig. Box counting. Top: A binary blood vessel structure after image processing. **Middle**: A log-log plot showing the number of boxes (N) required to cover the blood vessel in relation to box size (ϵ). The slope of the dashed red regression line is 1.315, corresponding to the estimated fractal dimension, D. **Bottom:** A log-log plot showing the mean λ -value in relation to decreasing box size (ϵ). The slope of the dashed red regression line is 0.454, corresponding to the estimated lacunarity of the blood vessel. (TIF)

S6 Fig. Template matching. A. The original colour image. **B.** After extracting the region of interest, the boosted image is thresholded by using the mean grey value. This leaves primarily blood vessel structures in the image. **C.** The circular convolution template. **D.** The result of the convolution of B and C (multiplication in the frequency domain). **E.** The result of thresholding image D on the 97.5th percentile of pixel value. **F.** The result of the template matching (E) superimposed on the original colour image (A). (TIF)

Acknowledgments

We would like to thank Roy Manyaira and the team at the BRIGHT (Bilharzia Research Improving Global Health Today) research clinic for their indefatigable work.

Author Contributions

Conceived and designed the experiments: SH HNGA EFK FA MT. Performed the experiments: SH HNGA EK EFK PP. Analyzed the data: SH FA. Contributed reagents/materials/analysis tools: SH HNGA EK EFK FA. Wrote the paper: SH HNGA EK PP TN MT MO EFK FA.

References

- Kjetland EF, Ndhlovu PD, Mduluza T, Gomo E, Gwanzura L, Mason PR, et al. Simple clinical manifestations of genital Schistosoma haematobium infection in rural Zimbabwean women. Am J Trop Med Hyg. 2005; 72: 311–319. PMID: <u>15772328</u>
- Renaud G, Devidas A, Develoux M, Lamothe F, Bianchi G. Prevalence of vaginal schistosomiasis caused by Schistosoma haematobium in an endemic village in Niger. Trans Roy Soc Trop Med Hyg. 1989; 83: 797. PMID: <u>2515636</u>
- Kjetland EF, Leutscher PDC, Ndhlovu PD. A review of female genital schistosomiasis. Trends Parasitol. 2012; 28: 58–65. doi: 10.1016/j.pt.2011.10.008 PMID: 22245065
- Norseth HM, Ndhlovu PD, Kleppa E, Randrianasolo BS, Jourdan PM, Roald B, et al. The Colposcopic Atlas of Schistosomiasis in the Lower Female Genital Tract. Based on Studies in Malawi, Zimbabwe, Madagascar and South Africa. PLoS Negl Trop Dis. 2014; 8: e3229. doi: <u>10.1371/journal.pntd.0003229</u> PMID: <u>25412334</u>
- Kjetland EF, Poggensee G, Helling-Giese G, Richter J, Sjaastad A, Chitsulo L, et al. Female genital schistosomiasis due to Schistosoma haematobium. Clinical and parasitological findings in women in rural Malawi. Acta Trop. 1996; 62: 239–255. PMID: 9028409
- Jourdan PM, Roald B, Poggensee G, Gundersen SG, Kjetland EF. Increased Vascularity in Cervicovaginal mucosa with Schistosoma haematobium infection. PLoS Negl Trop Dis. 2011; 5: e1170. doi: <u>10.</u> <u>1371/journal.pntd.0001170</u> PMID: <u>21666790</u>
- Jourdan PM, Randrianasolo BS, Feldmeier H, Chitsulo L, Ravoniarimbinina P, Roald B, et al. Pathologic mucosal blood vessels in active female genital schistosomiasis: new aspects of a neglected tropical disease. Int J Gynecol Pathol. 2013; 32: 137–40. doi: <u>10.1097/PGP.0b013e31824fe327</u> PMID: 23202777
- Colley DG, Bustinduy AL, Secor WE, King CH. Human schistosomiasis. Lancet. 2014; 383: 2253–64. doi: 10.1016/S0140-6736(13)61949-2 PMID: 24698483
- Kjetland EF, Ndhlovu PD, Gomo E, Mduluza T, Midzi N, Gwanzura L, et al. Association between genital schistosomiasis and HIV in rural Zimbabwean women. AIDS. Harare, Zimbabwe; 2006; 20: 593–600. PMID: 16470124
- Downs JA, Mguta C, Kaatano GM, Mitchell KB, Bang H, Simplice H, et al. Urogenital Schistosomiasis in Women of Reproductive Age in Tanzania's Lake Victoria Region. Am J Trop Med Hyg. 2011; 84: 364–369. doi: 10.4269/ajtmh.2011.10-0585 PMID: 21363971
- 11. Ndeffo Mbah ML, Poolman EM, Drain PK, Coffee MP, van der Werf MJ, Galvani AP. HIV and Schistosoma haematobium prevalences correlate in sub-Saharan Africa. Trop Med Int Health. 2013;00.
- 12. Kjetland EF, Norseth HM, Taylor M, Lillebø K, Kleppa E, Holmen SD, et al. Classification of the lesions observed in female genital schistosomiasis. Int J Gynaecol Obstet. Elsevier B.V.; 2014; 2–3.
- Wamala DS, Augustine K, others. A meta-analysis of telemedicine success in Africa. J Pathol Inform. Medknow Publications; 2013; 4: 6. doi: <u>10.4103/2153-3539.112686</u> PMID: <u>23858382</u>
- Holmen SD, Kjetland EF, Taylor M, Kleppa E, Lillebø K, Gundersen SG, et al. Colourimetric image analysis as a diagnostic tool in female genital schistosomiasis. Med Eng Phys. 2015; 37: 309–314. doi: <u>10.</u> <u>1016/j.medengphy.2014.12.007</u> PMID: <u>25630808</u>
- Holmen SD, Kleppa E, Lillebø K, Pillay P, van Lieshout L, Taylor M, et al. The first step towards diagnosing female genital schistosomiasis by computer image analysis. Am J Trop Med Hyg. 2015;
- Cremer ML, Peralta EI, Dheming SG, Jimenez ME, Davis-Dao C a, Alonzo T a, et al. Digital assessment of the reproductive tract versus colposcopy for directing biopsies in women with abnormal Pap smears. J Low Genit Tract Dis. 2010; 14: 5–10. doi: <u>10.1097/LGT.0b013e3181a02058</u> PMID: <u>20040830</u>
- 17. Ford C. Implementation of cervical cancer prevention services for HIV-infected women in Zambia: measuring program effectiveness. J Fam Plann Reprod Health Care. 2011; 37: 186.
- Hegertun IEA, Sulheim Gundersen KM, Kleppa E, Zulu SG, Gundersen SG, Taylor M, et al. S. haematobium as a Common Cause of Genital Morbidity in Girls: A Cross-sectional Study of Children in South Africa. Downs JA, editor. PLoS Negl Trop Dis. 2013; 7: e2104. doi: <u>10.1371/journal.pntd.0002104</u> PMID: <u>23556009</u>

- Morgas DT, Kvalsvig JD, Gundersen SG, Taylor M, Kjetland EF, Thomassen Morgas DE. Schistosomiasis and water-related practices in school girls in rural KwaZulu-Natal, South Africa. South African J Epidemiol Infect. 2010; 25: 30–33. Available: http://www.sajei.co.za/index.php/SAJEl/article/view/241
- Solomon D, Davey D, Kurman R, Moriarty A, O'Connor D, Prey M, et al. The 2001 Bethesda System: terminology for reporting results of cervical cytology. Jama. American Medical Association; 2002; 287: 2114–2119.
- Youssif AAHAR, Ghalwash AZ, Ghoneim AASAR. Optic disc detection from normalized digital fundus images by means of a vessels' direction matched filter. IEEE Trans Med Imaging. 2008; 27: 11–18. doi: 10.1109/TMI.2007.900326 PMID: 18270057
- Lalonde M, Beaulieu M, Gagnon L. Fast and robust optic disc detection using pyramidal decomposition and hausdorff-based template matching. IEEE Trans Med Imaging. 2001; 20: 1193–1200. PMID: <u>11700746</u>
- Reeves AA. Optimized Fast Hartley Transform for the MC68000 with Applications in Image Processing. MS Thesis, [Internet]. Thayer School of Engineering, Dartmouth College, Hanover, New Hampshire. 1990. Available: http://imagej.nih.gov/ij/download/docs/ImageFFT/thesis.pdf
- 24. Niblack W. An Introduction to Digital Image Processing. Prentice-Hall; 1986.
- 25. Chaki N, Shaikh SH, Saeed K. Exploring Image Binarization Techniques. Springer; 2014.
- Rasband WS. ImageJ [Internet]. Bethesda, Maryland, USA: U. S. National Institutes of Health; Available: <u>http://imagej.nih.gov/ij/</u>
- Landini G, Murray PI, Misson GP. Local connected fractal dimensions and lacunarity analyses of 60° fluorescein angiograms. Investig Ophthalmol Vis Sci. 1995; 36: 2749–2755.
- Doubal FN, MacGillivray TJ, Patton N, Dhillon B, Dennis MS, Wardlaw JM. Fractal analysis of retinal vessels suggests that a distinct vasculopathy causes lacunar stroke. Neurology. 2010; 74: 1102–1107. doi: 10.1212/WNL.0b013e3181d7d8b4 PMID: 20368631
- Smith TG, Lange GD, Marks WB. Fractal methods and results in cellular morphology—Dimensions, lacunarity and multifractals. J Neurosci Methods. 1996; 69: 123–136. PMID: <u>8946315</u>
- Vlachokosta A a., Asvestas P a., Gkrozou F, Lavasidis L, Matsopoulos GK, Paschopoulos M. Classification of hysteroscopical images using texture and vessel descriptors. Med Biol Eng Comput. 2013; 51: 859–867. doi: <u>10.1007/s11517-013-1058-1</u> PMID: <u>23504345</u>
- Boppart SA, Richards-Kortum R. Point-of-care and point-of-procedure optical imaging technologies for primary care and global health. Sci Transl Med. 2014; 6.
- Peeling RW, Holmes KK, Mabey D, Ronald a. Rapid tests for sexually transmitted infections (STIs): the way forward. Sex Transm Infect. 2006; 82 Suppl 5: v1–v6. PMID: <u>17151023</u>
- Drain PK, Hyle EP, Noubary F, Freedberg K a., Wilson D, Bishai WR, et al. Diagnostic point-of-care tests in resource-limited settings. Lancet Infect Dis. Elsevier Ltd; 2014; 14: 239–249. doi: <u>10.1016/</u> S1473-3099(13)70250-0 PMID: 24332389
- Chirenje ZM, Rusakaniko S, Kirumbi L, Ngwalle EW, Makuta-Tlebere P, Kaggwa S, et al. Situation analysis for cervical cancer diagnosis and treatment in east, central and southern African countries. Bull World Health Organ. 2001; 79: 127–132. PMID: 11242819
- Quinley KE, Gormley RH, Ratcliffe SJ, Shih T, Szep Z, Steiner A, et al. Use of mobile telemedicine for cervical cancer screening. J Telemed Telecare. 2011; 17: 203–209. doi: <u>10.1258/jtt.2011.101008</u> PMID: 21551217
- 36. Parham GP, Mwanahamuntu MH, Pfaendler KS, Sahasrabuddhe V V., Myung D, Mkumba G, et al. eC3-A Modern Telecommunications Matrix for Cervical Cancer Prevention in Zambia. J Low Genit Tract Dis. 2010; 14: 167–173. doi: <u>10.1097/LGT.0b013e3181cd6d5e</u> PMID: <u>20592550</u>
- Kleppa E, Holmen SD, Lillebø K, Kjetland EF, Gundersen SG, Taylor M, et al. Cervical ectopy: associations with sexually transmitted infections and HIV. A cross-sectional study of high school students in rural South Africa. Sex Transm Infect. 2014; 1–6.